Author's response to reviews

Title: Eligibility for interventions, co-occurrence and risk factors for unhealthy behaviours in patients consulting for routine primary care: results from the Pre-Empt study.

Authors:

Elizabeth Randell (randelle@cardiff.ac.uk)
Timothy Pickles (pickleste@cardiff.ac.uk)
Sharon A Simpson (sharon.simpson@glasgow.ac.uk)
Clio Spanou (clio.spanou@beds.ac.uk)
Jim McCambridge (jim.mccambridge@york.ac.uk)
Kerenza Hood (hoodk1@cardiff.ac.uk)
Christopher C Butler (Christopher.butler@phc.ox.ac.uk)

Version: 7
Date: 21 September 2015

Author's response to reviews:

21/09/2015

Dear Sir/Madam,

MS: 1430877615167927

Eligibility for interventions, co-occurrence and risk factors for unhealthy behaviours in patients consulting for routine primary care: results from the Pre-Empt study..

In response to Editor's comment:

** Please format the abstract, including the trial registration section, according to our Instructions for Authors: http://www.biomedcentral.com/bmcfampract/authors/instructions/researcharticle

The abstract has been amended and formatted according to the BMC Family Practice guidelines. I hope this covers all reviewers comments.

Yours Sincerely

Elizabeth Randell

Encl: Pre-Empt Screening paper v2.3